Author's response to reviews

Title: Calcitonin-producing well-differentiated neuroendocrine carcinoma (carcinoid tumor) of the urinary bladder: case report

Authors:

Massimo Mascolo (mmascol@libero.it)
Vincenzo Altieri (altieri@unina.it)
Chiara Mignogna (chiara.mignogna@libero.it)
Giorgio Napodano (giorgionapodano@virgilio.it)
Gaetano De Rosa (gaderosa@unina.it)
Luigi Insabato (insabato@unina.it)

Version: 3 Date: 18 July 2005

Author's response to reviews:

Calcitonin-producing well-differentiated neuroendocrine carcinoma (carcinoid tumor) of the urinary bladder: case report (M. Mascolo et al.)

Changes made according to the Editor

1. TITLE
Calcitonin-producing well-differentiated neuroendocrine carcinoma (carcinoid tumor) of the urinary bladder: case report
instead of
CALCITONIN-PRODUCING WELL-DIFFERENTIATED NEUROENDOCRINE CARCINOMA (CARCINOID TUMOR) OF THE URINARY BLADDER: CASE REPORT

2. QUALIFICATIONS
Massimo Mascolo1, mmascol@libero.it;
Vincenzo Altieri2, altieri@unina.it
Chiara Mignogna1, chiara.mignogna@libero.it;
Giorgio Napodano2, giorgionapodano@virgilio.it;
Gaetano De Rosa1, gaderosa@unina.it;
Luigi Insabato1, insabato@unina.it;

instead of
Massimo Mascolo, MD1, mmascol@libero.it;
Vincenzo Altieri MD2, altieri@unina.it
Chiara Mignogna, MD1, chiara.mignogna@libero.it;
Giorgio Napodano, MD2, giorgionapodano@virgilio.it;
Gaetano De Rosa, MD1, gaderosa@unina.it;
Luigi Insabato, MD, PhD1, insabato@unina.it;

3. AFFILIATIONS
1 Department of Biomorphological and Functional Sciences, Pathology section, University of naples
"Federico II", via S. Pansini 5, 80131, Naples, Italy;
2 Department of Urology, University of Naples "Federico ii", via S. Pansini 5, 80131, Naples, Italy.

Instead of
- University of Naples "Federico II", Naples, Italy:
1 Department of Biomorphological and Functional Sciences, Pathology Section;
2 Department of Urology.

4. SECTION HEADINGS
Abstract, Background, Case Presentation, Conclusion, References

instead of
ABSTRACT, BACKGROUND, CASE PRESENTATION, CONCLUSION, REFERENCES

5. ABSTRACT
Conclusion: this tumor shows specific clinical, macroscopical and histological features and must be considered in the differential diagnosis of bladder neoplasms

Instead of
Conclusion: We here add another case of carcinoid tumor of the urinary bladder to the existing literature, this tumor shows specific clinical, macroscopical and histological features and must be considered in the
Background: The occurrence of calcitonin-secreting primary carcinoid tumor of the urinary bladder is extremely rare.

Case presentation: The case of a 68-year-old male with carcinoid tumor arising in the urinary bladder is presented. Transurethral resection of a polypoid small tumor 0.4 cm in diameter was performed. Immunohistochemical study using neuroendocrine markers allowed a straightforward diagnosis of a low-grade neuroendocrine carcinoma (carcinoid tumor) of the urinary bladder. Immunohistochemistry demonstrated calcitonin immunoreactivity in the most of the tumor cells.

Conclusion: We here add another case of carcinoid tumor of the urinary bladder to the existing literature, this tumor shows specific clinical, macroscopical and histological features and must be considered in the differential diagnosis of bladder neoplasms.

Instead of

Background: The occurrence of calcitonin-secreting primary carcinoid tumor of the urinary bladder is extremely rare.

Case presentation: The case of a 68-year-old male with carcinoid tumor arising in the urinary bladder is presented. Transurethral resection of a polypoid small tumor 0.4 cm in diameter was performed. Immunohistochemical study using neuroendocrine markers allowed a straightforward diagnosis of a low-grade neuroendocrine carcinoma (carcinoid tumor) of the urinary bladder. Immunohistochemistry demonstrated calcitonin immunoreactivity in the most of the tumor cells.

Conclusion: We here add another case of carcinoid tumor of the urinary bladder to the existing literature, this tumor shows specific clinical, macroscopical and histological features and must be considered in the differential diagnosis of bladder neoplasms.

7. REFERENCES
References are cited in the text using consecutive numbers in square brackets.

8. CONCLUSION
Conclusions

9. PAGE BREAKS
I remove page breaks from the manuscript after the title page.

10. COMPETING INTEREST
The authors declare that they have no competing interests.

11. AUTHORS’ CONTRIBUTIONS
MM participated to write the manuscript.
VA operated the patient and partecipated to draft the manuscript.
CM carried out immunohistochemical study.
GN operated the patient and partecipated to draft the manuscript.
GDR partecipated in the design of the study.
LI partecipated in the design of the study.

12. REFERENCES
I remove the sentence indicating the type of reference form the reference list.